In answer to your mail of 17th July:

- Please give the full name and affiliation of the two ethics committees that approved the study.

In our manuscript we mentioned “Permission for the recruitment procedures was obtained from the relevant Research Ethics Committee of both hospitals”.

In addition we mention the names of the ethics committees:

- Institutional Review Board of the Radboud university medical center, Nijmegen the Netherlands:
  Permission was given with number ABR nr NL.21205.091.07 (10 June 2008)
  Names of the members of the Institutional Review Board:
  Dr. Mr. F.M. van Agt, head of the office
  Dr. F.Th.M. Huysmans, Vice Chairman.
  Dr. M.J.J. Prick, Vice Chairman.
  Mrs. L.M. Damen MSc, Secretary.
  Mrs. dr. ir. M Verlaan, Secretary.

- Institutional Review Board of the Sint Elisabeth Hospital, Tilburg the Netherlands:
  Permission was given with protocol number 1106 (25 March 2011).

- In the abstract (in the manuscript and in the submission system) please change the section heading Findings to Results, in accordance with our guidelines.

We changed the headings in accordance to your guidelines.